

Dear Reviewer,

Date: April 20, 2024

Thank you for your comments regarding our manuscript entitled **Importance of risk assessment, endoscopic hemostasis, and recent advancements in the management of acute non-variceal upper gastrointestinal bleeding** (Manuscript NO 94073). Your review points are all addressed point-by-point in the section below:

**Reviewer #1:** Good quality review. Needs some English language polishing. I would have liked to be more precise in the benefit and usage fo scoring system.

- Got the paper reviewed by a native English-speaking expert and carried out changes in the Title and grammar to polish the language.
- Explained the benefits of using scoring systems on page 5 and added a new score (INBS) on page 6 to illustrate their usage.

We hope the new manuscript is deemed acceptable by the Journal for publication.

Yours sincerely,  
Arkadeep Dhali